Author’s response to reviews

Title: Structural and cognitive deficits in chronic carbon monoxide intoxication: a voxel-based morphometry study

Authors:

Hsiu-Ling Chen (suring.tw@gmail.com)
Pei-Chin Chen (pachin1117@gmail.com)
Cheng-Hsien Lu (chlu99@adm.cgmh.org.tw)
Nai-Wen Hsu (almhsu36@hotmail.com)
Kun-Hsien Chou (dargonchow@gmail.com)
Ching-Po Lin (chingpolin@gmail.com)
Re-Wen Wu (ray4595@adm.cgmh.org.tw)
Shau-Hsuan Li (ma4844@adm.cgmh.org.tw)
Yu-Fan Cheng (cheng.yufan@msa.hinet.net)
Wei-Che Lin (u64lin@yahoo.com.tw)

Version: 7 Date: 26 September 2013

Author’s response to reviews: see over
Dear The BMC Neurology Editorial Team,

Please find enclosed our revised manuscript entitled, “Structural and cognitive deficits in chronic carbon monoxide intoxication: a voxel-based morphometry study” (MS: 1906842155811333), which has been modified in response to your comments. We have revised the abstract and ensured that the revised manuscript conforms to the journal style.

Editorial requirement:
1. Please could you structure your abstract according to the guidelines provided at this page: http://www.biomedcentral.com/info/ifora/abstracts

We have made the revision in the manuscript accordingly.
The abstracted has been re-structured as follows: Background, Methods, Results, and Conclusions.

We thank you very much for your consideration.

Yours sincerely and on behalf of all authors,

Hsiu-Ling Chen, MD

Correspondence to:
Wei-Che Lin, MD
Department of Diagnostic Radiology, Kaohsiung Chang Gung Memorial Hospital
123 Ta-Pei Road, Niao-Sung, Kaohsiung, 83305, Taiwan
Tel: +886-7-7317123 Ext. 3027
Fax: +886-7-7317123 Ext. 2523
Email: u64lin@yahoo.com.tw